Author's response to reviews

Title: Prevalence and correlates of physical disability and functional limitation among community dwelling older people in rural Malaysia, a middle income country

Authors:

Noran N Hairi (noran@um.edu.my)
Awang Bulgiba (awang@um.edu.my)
Robert G Cumming (robert.cumming@sydney.edu.au)
Vasi Naganathan (vasi.naganathan@sydney.edu.au)
Izzuna M Ghazali (drizzuna@moh.gov.my)

Version: 2 Date: 3 August 2010

Author's response to reviews: see over
29th July 2010

Dr. Matthew Kaiser,
Scientific Editor
BMC-series Journals

MS: 3377245363917778
Prevalence and correlates of physical disability and functional limitation among older people in Malaysia, a middle income country

Dear Dr Kaiser,

Thank you for the opportunity to revise our manuscript. We are grateful to the two reviewers for their time and valuable comments, and we have found their comments most helpful.

Revisions made in the manuscript are systematically indicated in the table below.

We appreciate the opportunity to address these issues and believe that the paper is much improved as a consequence.

Yours sincerely,

Dr. Noran Naqiah Hairi